Author’s response to reviews

Title: A Review of Autobiographical Memory Studies on Patients with Schizophrenia Spectrum Disorders

Authors:

Yujia Zhang (yujia.zhang@monash.edu)

Sara Kuhn (sara.kuhn@UND.edu)

Laura Jobson (Laura.jobson@monash.edu)

Shamsul Haque (shamsul@monash.edu)

Version: 4 Date: 24 Oct 2019

Author’s response to reviews:

1. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

Authors’ response: A list of abbreviations has been included in the Declarations (page 35-36).

2. In the "Availability of Data and Materials" section, please detail where the raw data supporting your findings can be found (including information found in any supplementary files). If the raw data is publicly available or can be requested, please state that this is the case, and where the data can be found/requested from. Alternatively, if you do not wish to/cannot share your data, please state (in the ‘Availability of data and Materials’ section) that data will not be shared, and state the reason.

Authors’ response: We have included the statement “Data sharing is not applicable to this article as no datasets were generated or analysed during the current study” in the "Availability of Data and Materials" section (page 36).

3. All sources of funding for the research reported should be declared. The role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript should be declared. If no funding was obtained for this study, please state so in this section.
We appreciate that a statement providing funding information may already be present. Please move/copy the statement to the Funding sub-section instead, and include any additional relevant information.

Authors’ response: Thanks for this suggestion. We have declared the source of funding in the Declarations (page 37).

4. Please move the statement 'Yujia Zhang is supported by a Higher Degree by Research Scholarship from Monash University Malaysia.' into the competing interests statement.

Authors’ response: Thank you very much for this. We have now moved the statement into the competing interests statement (page 36-37).

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Authors’ response: We have followed this advice; the clean version of the manuscript has been submitted. The tables and supplementary files have been submitted separately.